Author's response to reviews

Title: Why do hypertensive patients of African ancestry respond better to calcium blockers and diuretics than to ACE inhibitors and beta-adrenergic blockers? A systematic review

Authors:

LIZZY M BREWSTER (L.M.BREWSTER@AMC.UVA.NL)
YACKOOB K SEEDAT (seedaty1@gmail.com)

Version: 10 Date: 13 March 2013

Author's response to reviews: see over
Dear Madam,

Thank you for the reply of the reviewer and your kind acceptance of our work. Please find enclosed a list of final edits. The manuscripts, a clean version and one with tracked changes, have been uploaded.

Yours faithfully,

On behalf of Prof. Y.K. Seedat,

Lizzy Brewster, MD, PhD
Hypertension Group
Depts. of Internal and Vascular Medicine, F4-222
Academic Medical Center, University of Amsterdam
Meibergdreef 9, 1105 AZ Amsterdam
The Netherlands
Tel: *31-20-566 3039; Fax: *31-84-872 3752
E-mail: l.m.brewster@amc.uva.nl
Web: www.amc.nl/hypertension
Comments of the Editor and Reviewer

Reviewers

Referee 1 is satisfied with the revisions.

Editorial Requests

1. We provided the email addresses of all authors

2. We provided authors’ contributions in the journal format

3. We included an abbreviation section, with small abbreviation edits (tracked)

4. We checked whether our revised manuscript conforms to the journal style, such that the files are now correctly formatted. (We did not use track changes for these formatting edits, as the text did not change, only the formatting)